Author's response to reviews

Title: Study protocol: Follow-up home visits with nutrition: a randomised controlled trial

Authors:

Anne Marie Beck (abec0018@heh.regionh.dk)
Stine Kjær (stinekjaer@live.dk)
Birthe S Hansen (bsha@life.ku.dk)
Rikke L Storm (rikkelunau@hotmail.com)
Kirsten Thal-Jantzen (kthaljantzen@msn.com)

Version: 4 Date: 19 December 2011

Author’s response to reviews:

We have made the following changes

Author list: The list of authors in the manuscript is written exactly as they are 'First name Initial Last name'

Authors contributions: We have removed from the title page and placed it towards the end of the manuscript text, directly preceding the Acknowledgements

References: We are now sure that references are cited in numerical order and reference 20 is now cited within the main text.

References: We have provided a functioning link to website www.sf-36.org

Email address: We have included email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Affiliations: We have numbered the affiliations 1 and 2, and linked the numbers to the author list

Abstract: We have changed the title "Methods" to "Methods/Design" in the manuscript copy of the abstract

Methods/Design: We have changed the title "Methods" to "Methods/Design"

Typography: We have checked our manuscript for any typographical errors and made some final corrections.